Author's response to reviews

**Title:** Assessment of knowledge, attitude and practice and associated factors towards palliative care among nurses working in selected hospitals, Addis Ababa, Ethiopia

**Authors:**

Hiwot Kassa (hiwiworeta@gmail.com)
Rajalakshmi Murugan (fmzewdu@gamil.com)
Fissiha Zewdu (atse2000m@yahoo.com)
Mignote Hailu (elatman@gmail.com)
Desalegn Woldeyohannes (desalegn01@gmail.com)

**Version:** 4  
**Date:** 26 December 2013

**Author's response to reviews:** see over
Dear BMC Public Health Journal Editors,

As per reviewers request, we revised the manuscript to improve its quality.

To reviewers one and two

We let the whole manuscript, including the abstract section, got revised for its English language quality by native English language speaker and we made corrections accordingly.

Point by point response to major compulsory revisions, Version: 3, Martin Weber

<table>
<thead>
<tr>
<th></th>
<th>operational definition: what about scores between 50 and 59% in the FATCOD – Scale?</th>
<th>Unfavorable attitude = &lt; 50% of the total score of (FATCOD) Scale. Favorable attitude = ≥ 50 of the total score of (FATCOD) Scale. Correction has been made and incorporated in the text.</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>Belonging to a specific religious community could strongly influence the attitude Towards Palliative Care. If this socio demographic item was not assessed, this should be noted in the methods section.</td>
<td>The reason for not including religion in the socio demographic section was during literature review most studies done in the area didn’t illustrate association between PC and religion. To make it clear we mentioned all socio demographic variables included in the Methodology section (Data collection instrument and technique).</td>
</tr>
<tr>
<td>2</td>
<td>Answer for question number 5 of the cover letter do not have a correlate in the revised manuscript</td>
<td>• Answer for this question included in the methodology section: Data collection instrument and technique</td>
</tr>
</tbody>
</table>